Author's response to reviews

Title: Diffusion of good practices of care and decline of the association with case volume: the example of breast conserving surgery

Authors:

Ugo Fedeli (ugo.fedeli@ulssasolo.ven.it)
Natalia Alba (natalia.alba@ulssasolo.ven.it)
Elena Schievano (elena.schievano@ulssasolo.ven.it)
Cristiana Visentin (cristiana.visentin@ulssasolo.ven.it)
Rosalba Rosato (rosalba.rosato@cpo.it)
Manuel Zorzi (manuel.zorzi@istitutoncologicoveneto.it)
Giancarlo Ruscitti (segr.sanita@regione.veneto.it)
Paolo Spolaore (paolo.spolaore@ulssasolo.ven.it)

Version: 3 Date: 17 October 2007

Author's response to reviews:

Dear Editor,

we've performed the requested formatting changes:

1) The list of authors in the manuscript is now identical (same order) to that in the submission system

2) Each affiliation is on one line

3) Section headings are in the correct format

4) A website address has been removed from the text and added in the reference list

5) The sentence about competing interests has been included in the Competing interest section

Ugo Fedeli